Author's response to reviews

Title: Female genital mutilation of a karyotypic male presenting as a female with delayed puberty

Authors:

Mona Ellaithi (Ellaithi_mon@yahoo.com)
Therese Nilsson (Therese.Nilsson@med.lu.se)
David Gisselsson (David.Gisselsson@med.lu.se)
Atif Elagib (Atifelaqib@yahoo.com)
Hamid Eltigani (Heeltigani@yahoo.com)
Imad Fadi-Elmula (Fadi-elmula@neelain.info)

Version: 6 Date: 27 February 2006

Author's response to reviews:

Dear Editor,

Please find enclosed the revised version of our manuscript Female Genital Mutilation of a karyotypic male presenting as a female with delayed puberty. The manuscript language has been revised by an English language professor in university of Khartoum. After that we have rechecked the manuscript again in order to check if the language changes have affected the scientific information. However, the language of the manuscript was very well written as the confirmed by the English teacher. We have also checked the manuscript final format.

Thank you for your care,
Mona Ellaithi
Corresponding author